Author's response to reviews

Title: A case report of the effect of the dual endothelin receptor antagonist, bosentan, on untreatable ulcers in a diabetic patient

Authors:

Fatima Álvarez Reyes Dr (falvrey@gmail.com)
Cristina Luna Gómez Dr (cristinaluna76@gmail.com)
Manuel Brito Suárez Dr (mbrisua@gobiernodecanarias.org)

Version: 5 Date: 4 March 2011

Author's response to reviews:

Dear Deputy Editor

In response to your request to provide more information in the conflict of interest section the following has been added to the manuscript:

"""We confirm that we have no financial or non-financial competing interests. We have not received funding in the form of a grant or received fees for the writing of this article. Actelion Spain paid for medical writing assistance as we wanted the opportunity for our case to be published in an International journal. As non-native English speakers the process is difficult and so we accepted the support of Elements Communications Ltd, funded by Actelion Spain, which we have duly acknowledged as per Good Publications Practice."

Yours sincerely